FDA125 Advanced Parallel Programming

Feedback form — Programming assignment

Please check as appropriate:

I installed / used the Fork compiler and PRAM simulator 2 Yes 2 No, because I already solved the programming exercise 2 Yes 2 No 2 partially I found the assignment 2 too easy 2 too difficult 2 just the right degree of difficulty I looked at some of the example programs before I started: 2 Yes 2 No I used the course book Practical PRAM Programming as a guide to programming in Fork: 2 Yes 2 No I found the trace file visualizer (trv) useful (a) to identify performance bottlenecks 2 Yes 2 No (b) to understand the structure of the parallel computation 2 Yes 2 No (c) to debug my program 2 Yes 2 No (d) to I learned something by doing the programming assignment (a) about the theoretical concepts of parallel programming 2 Yes 2 No (b) about implementation problems 2 Yes 2 No I think the programming assignment is a useful complement of the other parts of the course

2 Yes 2 No, because

I recommend the following technical improvements: 2 2 I recommend the following improvements in the documentation: 2 2 Further comments: 2 2 2 Thank you for your comments!